Dovepress

Joint Effects of Plasma Homocysteine Concentration and Traditional Cardiovascular **Risk Factors on the Risk of New-Onset Peripheral** Arterial Disease

This article was published in the following Dove Press journal: Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy

Purpose: Hyperhomocysteinemia is an independent risk factor for cardio- and cerebrovascular diseases. However, the relationship between plasma homocysteine (Hcy) concentration and peripheral arterial disease (PAD) has not been completely characterized. The aim of the present study was to determine the relationship between plasma Hcy concentration and new-onset PAD and to assess the effects of combinations of Hcy and traditional cardiovascular risk factors.

Patients and Methods: We conducted a prospective community-based cohort study of 3119 Chinese participants who did not have PAD at baseline, with a median follow-up period of 2.30 years. We used multivariate logistic regression models to evaluate the relationship between high Hcy (≥10µmol/L) and new-onset PAD. The effects of combinations of high Hey and traditional cardiovascular risk factors were assessed using logistic regression analysis.

Results: After adjustment for 14 covariates, high Hcy concentration was significantly associated with new-onset PAD (odds ratio [OR]=2.08, 95% confidence interval [CI]: 1.-08-4.03, P=0.030). Smokers with high Hcy concentration were substantially more likely to have new-onset PAD than non-smokers with normal Hcy concentration (OR=4.44, 95% CI: 1.77-11.12, P=0.001). The effect of diabetes on PAD became significant when present in combination with high Hcy concentration (OR=3.67, 95% CI: 1.25-10.80, P=0.018). Participants with both elevated Hcy levels and older age had the highest risk of new-onset PAD (OR=4.28, 95% CI: 1.83–10.01, P<0.001). With regard to the joint effect of Hcy and hypertension, dyslipidemia or sex, there was also a trend towards increased risk across four different groups (P for trend=0.026, 0.035, 0.016, respectively).

Conclusion: High plasma Hey concentration independently predicts the incidence of PAD. Furthermore, there is a joint effect of high Hcy concentration and traditional cardiovascular risk factors such as smoking, diabetes and aging on the incidence of PAD.

Keywords: peripheral arterial disease, atherosclerosis, hyperhomocysteinemia, cohort study, community-based population

Plain Language Summary

Why Was the Study Done?

- Homocysteine (Hcy) is involved in various pathological processes, such as endothelial dysfunction, oxidative stress, and vascular remodeling.
- However, the relationship between high Hcy concentration and peripheral arterial disease (PAD) remains unclear.

Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy 2020:13 3383-3393 CC 0 S C2020 Liu et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms.php you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs 4.2 and 5 of our Terms (https://www.dovepress.com/terms.php).

Danmei He^{1,2} Jiahui Liu^{1,2} Yuxi Li^{1,2} Yong Huo Jianping Li Yan Zhang^{1,2} ¹Department of Cardiology, Peking University First Hospital, Beijing, People's Republic of China; ²Institute of

Cardiovascular Disease, Peking

Republic of China

University First Hospital, Beijing, People's

Mengyuan Liu^{1,2}

Yimeng Jiang^{1,2}

Pengfei Sun^{1,2}

Bo Liu^{1,2}

Jia Jia^{1,2}

Fangfang Fan (D^{1,2}

Correspondence: Yan Zhang; Jianping Li Tel +86 10 83575262; +86 10 83575728 Fax +86 10 66551383 Email drzhy1108@163.com; lijianping03455@pkufh.com



3383

• Moreover, it is also unclear whether Hcy and traditional cardiovascular risk factors have a joint effect on PAD.

What Did the Researchers Do and Find?

- We conducted a prospective community-based cohort study of 3119 Chinese participants who did not have PAD at baseline, with a median follow-up period of 2.30 years.
- In this prospective cohort study, high plasma Hcy concentration was significantly associated with new-onset PAD.
- Moreover, there was a joint effect of high plasma Hcy concentration and traditional cardiovascular risk factors such as smoking, diabetes and aging on the incidence of PAD.

What Do These Results Mean?

- Our study provides new insights into the PAD risk associated with combinations of high Hey concentration and traditional risk factors.
- Early screening and appropriate treatments are essential among subjects with traditional risk factors accompanying with hyperhomocysteinemia.

Introduction

Peripheral artery disease (PAD) is the term used to describe atherosclerotic occlusion or stenosis of non-cardiac and non-intracranial arteries,^{1,2} and it is the third most common manifestation of atherosclerosis after coronary heart disease and stroke.³ Globally, more than 200 million people have PAD,⁴ and its prevalence is expected to increase as the population ages.⁵ PAD is strongly associated with a higher risk of cardiovascular events, and it is estimated that there was a 30% increase in the incidences of mortality and disability due to PAD between 2005 and 2015; therefore, it represents a major public health concern.^{6,7} A diagnosis of PAD is made when patients have an ankle brachial index (ABI) ≤0.90². PAD shares risk factors with other atherosclerotic vascular diseases, with smoking and diabetes being the most potent risk factors.³ However, evidence for specific risk factors for PAD is less robust than for other atherosclerotic vascular diseases.⁸

In 1969, McCully firstly proposed that the total plasma concentration of homocysteine (Hcy) is high in 15–40% of patients with coronary or cerebral arterial disease, or in individuals with thickened carotid arteries.^{9,10} Furthermore, previous studies have shown that hyperhomocysteinemia is an independent and modifiable risk factor for the development of atherosclerosis.¹¹ Because PAD is one of the manifestations of systemic atherosclerosis, the potential association between Hcy and PAD has been a subject of great interest. However,

studies of the relationship between Hcy and PAD have yielded inconsistent results,^{12–20} and it requires further characterization in Chinese populations. This is important because there are ethnic disparities in the genetic variations in Hcy metabolic enzymes and lifestyle diversity affecting nutritional status.⁸ Moreover, the effects of combinations of hyperhomocysteinemia with traditional cardiovascular risk factors on the prevalence of PAD are also worth exploring. Therefore, the present study was designed to determine the relationship between plasma Hcy concentration and PAD in a Chinese population and to investigate the effects of combinations of hyperhomocysteinemia and traditional cardiovascular risk factors, including hypertension, diabetes, dyslipidemia, smoking, body mass index (BMI), age and sex.

Materials and Methods Study Population

Participants were recruited from a cohort study of atherosclerosis that included 5962 residents of Beijing, China who were >40 years old, and which was conducted between December 2011 and April 2012. The details of the study procedures have been described previously.²¹ Participants were invited for a follow-up examination that took place between May 2014 and July 2014, of whom 3823 (64.1%) attended. The non-responders did not differ substantially from the responders with regard to their baseline characteristics, which were described in detail previously.²¹ Patients who had an ABI ≤0.9 at baseline, did not have baseline or follow-up ABI values recorded, or did not have a baseline plasma Hcy concentration recorded were excluded. Consequently, a total of 3119 participants were included in the final analysis (Figure S1). This study was approved by the ethics committee of Peking University, and written informed consent was obtained from each participant. The study was conducted in accordance with the ethical guidelines of the Declaration of Helsinki.

Data Collection

Baseline data were collected by trained research coordinators using a standard operating procedure. All the participants were interviewed using a standardized questionnaire to collect basic information, including regarding sociodemographic status, lifestyle, and medical history. At baseline, current smoking was defined as smoking at least one cigarette per day for at least half a year and current drinking was defined as drinking alcohol once or more per week for at least half a year. Anthropometric measurements were also made at baseline according to a standard operating procedure.²¹ BMI was calculated as body mass (kg)/height² (m²). Brachial blood pressure (BP) was measured in the right arm of each participant after they had been seated for 5 min using an Omron HEM-7117 electronic sphygmomanometer (Kyoto, Japan), a standard method of calibration, and appropriately sized cuffs. Triplicate measurements were made, with >1 min between successive readings, and each patient's systolic BP (SBP) and diastolic BP (DBP) was calculated as the mean of three consecutive measurements. At baseline, hypertension was defined as a self-reported history of hypertension, the use of any antihypertensive drug, an SBP ≥140mmHg, or a DBP ≥90mmHg. Diabetes was defined as a self-reported history, the use of any hypoglycemic drug, a fasting blood glucose (FBG) concentration \geq 7mmol/L, or a 2-h blood glucose concentration during oral testing \geq 11.1mmol/L. glucose tolerance (OGTT) Dyslipidemia was self-reported or defined as the use of any lipid-lowering drug; or concentrations of triglyceride (TG) ≥ 1.7 mmol/L (150 mg/dl), total cholesterol (TC) \geq 5.18mmol/L (200 mg/dl),low-density lipoproteincholesterol (LDL-C) ≥3.37mmol/l (130mg/dl), or highdensity lipoprotein-cholesterol (HDL-C) <1.04mmol/L (40mg/dl). Cardiovascular disease (CVD) was defined as a self-reported history of coronary heart disease, stroke, or transient ischemic attack (TIA). To be consistent with previous reports,²²⁻²⁵ we defined hyperhomocysteinemia as a total plasma Hcy concentration $\geq 10 \mu mol/L$.

Ankle Brachial Index

After having been in a supine position for at least 5 min, each participant's SBP was measured in both arms (the brachial artery) and ankles (the posterior tibial artery) simultaneously using a BP-203RPE III machine (Omron Healthcare). The machine calculated the ABI results automatically, and the left and right ABI values were defined as the ankle SBP on each side, divided by the highest brachial SBP. The overall ABI value was taken to be the lower of the two values obtained. ABI was measured twice during the study: once at baseline and once at the follow-up appointment in May–July 2014. New-onset PAD was defined as the presence of an ABI value ≤0.9 at the follow-up appointment in 2014.

Blood Sample Collection and Laboratory Methods

After an overnight fast of at least 12 h, a venous blood sample was obtained from the forearm of each participant at

baseline. Plasma samples were separated within 30 min of collection and were stored at -80°C. Serum samples were used for the measurement of FBG, 2-h blood glucose during a standard 75-g OGTT, TC, LDL-C, HDL-C, TG, and creatinine. Plasma Hcy concentration was measured enzymatically using an automated biochemical analyzer (Beckman Coulter AU480, California, United States). The principle of the method has been described in detail previously.²⁶ Other laboratory parameters were measured at baseline using a Roche C8000 Automatic Analyzer (Basel, Switzerland).

Statistical Analysis

Continuous variables are reported as median (interquartile range, IOR) and categorical variables are reported as numbers and percentages. The characteristics of the participants and the differences among the participants, classified according to their Hcy concentration (<10 or \geq 10µmol/L), were compared using Student's t-test for continuous variables or Pearson's chi-square test for categorical variables. For non-normally-distributed data, the Kruskal-Wallis Rank Test was used for continuous variables. Univariate and multivariate logistic regression models were used to determine the relationships of plasma Hcy (as a binary variable) and other variables with the development of newonset PAD. Two sets of multivariable models were used: Model 1, which was adjusted for baseline ABI, sex, and age; and Model 2, which was further adjusted for BMI, estimated glomerular filtration rate (eGFR) subgroup $(eGFR \ge 90 mL/min/1.73 m^2, 60 \le eGFR < 90 mL/min/1.73 m^2)$ or eGFR<60mL/min/1.73m²), current smoking, current drinking, hypertension, diabetes, dyslipidemia, cardiovascular disease, use of an antihypertensive agent, use of a lipidlowering agent, and use of a hypoglycemic agent (When the odds ratio of new-onset PAD associated with a certain factor is assessed, this factor is not adjusted). To determine the effects of combinations of hyperhomocysteinemia and hypertension on the risk of PAD, we divided the subjects into four different groups according to Hcy (≥10µmol/L/ <10µmol/L) and hypertension (yes/no), and compared their risk of PAD to the group with no hypertension and with Hcy concentration <10µmol/L. P for trend represents the linear trend across these four different groups. The joint effects of Hcy and diabetes, smoking, or dyslipidemia, BMI, age, sex were assessed in the same way. Interaction and stratified analyses were performed according to chronic disease histories (hypertension, diabetes, or dyslipidemia), smoking status, BMI, age and sex. All analyses were performed using Empower(R) (www.empowerstats.com, X&Y

solutions Boston, MA, USA) and R (<u>http://www.R-project.</u> <u>org</u>). P < 0.05 (two-sided) was considered to represent statistical significance.

Results

Baseline Patient Characteristics

The baseline characteristics of all the participants, both overall and stratified according to Hcy concentration, are shown in Table 1. The participants were 56 (51-62) years old; 37.10% were male, and their BMI was 25.87 (23.82–28.03) kg/m². Hypertension was present in 49.86% (n = 1555) of the participants, diabetes in 23.85% (n = 744), CVD in 12.63% (n = 394), dyslipidemia in 72.14% (n = 2250), and current tobacco use in 18.95% (n = 591). At the time of enrollment, the median (IQR) Hey concentration was 11.91 (9.97-14.71) µmol/L. Participants with an Hcy concentration $\geq 10 \mu mol/L$ had a significantly higher prevalence of hypertension and CVD, lower eGFR, and were more likely to be taking an antihypertensive drug. Current tobacco and alcohol use were also more frequent in the subgroup of participants with higher plasma Hcy concentration.

able i basenne enalacteristics of the rai departe	Table I	Baseline	Characteristics	of the	Participants
---------------------------------------------------	---------	-----------------	-----------------	--------	--------------

Factors Associated with New-Onset PAD

Over a median follow-up period of 2.30 years (25th-75th percentile: 2.30-2.40 years), the incidence of PAD was 3.14% (n = 98). Table 2 displays the results of the multivariate regression analysis of the relationships of Hcy and traditional cardiovascular risk factors with new-onset PAD. An Hey concentration $\geq 10 \mu mol/L$ was associated with a higher risk of new-onset PAD in Model 1 (odds ratio (OR) = 2.21, 95% confidence interval (CI): 1.16-4.21, P = 0.016). This relationship remained significant (OR = 2.08, 95% CI: 1.08-4.03, P = 0.030) in Model 2. Current smokers were more likely to experience new-onset PAD (OR = 2.20, 95% CI: 1.14–4.22, P = 0.018 in Model 2). Diabetes and hypertension seemed to be a risk factor for PAD in univariate regression model (OR = 1.99, 95% CI: 1.31–3.01, P = 0.001, and OR = 1.54, 95% CI: 1.02–2.33, P = 0.039, respectively), but its influence did not remain significant in Model 1 and 2. The association between dyslipidemia and PAD was not significant (P =0.060 in univariate regression model). Aging was associated with a higher risk of new-onset PAD (OR = 2.15, 95% CI: 1.30–3.55, P = 0.003 in full adjusted Model). The association between BMI categories, or sex and PAD was

	All (n=3119)	Hcy <l0µmol l<br="">(n=792)</l0µmol>	Hcy≥I0µmol/L (n=2327)	P value
ABI, median (IQR) Age, median (IQR), years Male, n (%) BMI, median (IOR), kg/m ²	1.11 (1.06–1.16) 56 (51–62) 1157 (37.10) 25.87 (23.82–28.03)	1.10 (1.05–1.15) 53 (48–57) 83 (10.48) 25.78 (23.53–27.87)	1.12 (1.06–1.17) 57 (52–63) 1074 (46.15) 25.89 (23.92–28.08)	<0.001 <0.001 <0.001 0.108
eGFR subgroup, n (%) eGFR ≥ 90mL/min/1.73m ² 60 ≤eGFR<90mL/min/1.73m ² eGFR<60mL/min/1.73m ² Current smoking, n (%) Current drinking, n (%)	2166 (69.49) 893 (28.65) 58 (1.86) 591 (18.95) 721 (23.12)	699 (88.26) 92 (11.62) 1 (0.13) 54 (6.82) 90 (11.36)	1467 (63.10) 801 (34.45) 57 (2.45) 537 (23.08) 631 (27.12)	<0.001 <0.001 <0.001
Prevalence of disease Hypertension, n (%) Diabetes, n (%) Dyslipidemia, n (%) CVD, n (%)	1555 (49.86) 744 (23.85) 2250 (72.14) 394 (12.63)	342 (43.18) 179(22.60) 571 (72.10) 77 (9.72)	1213 (52.13) 565 (24.28) 1679 (72.15) 317 (13.62)	<0.001 0.338 0.975 0.004
Medication Antihypertensive drugs, n (%) Hypoglycemic drugs, n (%) Lipid-lowering drug, n (%)	990 (31.94) 310 (9.97) 322 (10.43)	221 (28.08) 78 (9.87) 86 (11.04)	769 (33.25) 232 (10.00) 236 (10.23)	0.007 0.916 0.523

Abbreviations: Hcy, homocysteine; ABI, ankle brachial index; BMI, body mass index; eGFR, estimated glomerular filtration rate; CVD, cardiovascular disease; IQR, interquartile range.

Factor	Crude		Model I ^a		Model 2 ^b		
	OR (95% CI)	P value	OR (95% CI)	P value	OR (95% CI)	P value	
Hcy≥10µmol/L	2.49 (1.36–4.59)	0.003	2.21 (1.16–4.21)	0.016	2.08 (1.08-4.03)	0.030	
Hypertension	1.54 (1.02–2.33)	0.039	1.34 (0.86–2.11)	0.197	1.19 (0.63–2.25)	0.592	
Diabetes	1.99 (1.31–3.01)	0.001	1.45 (0.93–2.27)	0.099	1.28 (0.70–2.32)	0.424	
Dyslipidemia	1.63 (0.98-2.70)	0.060	1.35 (0.80-2.27)	0.264	1.12 (0.65–1.93)	0.688	
Current smoking	1.32 (0.82–2.13)	0.247	2.11 (1.14–3.90)	0.017	2.20 (1.14-4.22)	0.018	
BMI, kg/m ²							
<24	Ref.		Ref.		Ref.		
≥24, <28	0.88 (0.54-1.46)	0.630	0.88 (0.53-1.47)	0.624	0.87 (0.51–1.49)	0.609	
≥28	1.28 (0.76-2.18)	0.356	0.99 (0.57–1.73)	0.982	0.95 (0.53–1.71)	0.868	
Age≥60 years	2.66 (1.78–3.99)	<0.001	3.22 (2.12-4.88)	<0.001	2.15 (1.30-3.55)	0.003	
Female	1.17 (0.76–1.79)	0.477	0.97 (0.62–1.52)	0.907	1.36 (0.74–2.50)	0.322	

Table 2 Odds Ratio for New-Onset PAD Associated with High Hcy Concentration and Traditional Cardiovascular Risk Factors

Notes: ^aModel 1: adjusted for baseline ABI, sex, and age. ^bModel 2: adjusted for baseline ABI, sex, age, body mass index, estimated glomerular filtration rate subgroup, current smoking and drinking, hypertension, diabetes, dyslipidemia, cardiovascular disease, and the use of antihypertensive, lipid-lowering, or hypoglycemic agents. (When the odds ratio of new-onset PAD associated with a certain factor is assessed, this factor is not adjusted. For example, dyslipidemia is not adjusted when the odds ratio of new-onset PAD associated with dyslipidemia is assessed).

Abbreviations: Hcy, homocysteine; OR, odds ratio; Cl, confidence interval; PAD, peripheral artery disease; ABI, ankle brachial index; BMI, body mass index.

not significant. Interaction analysis showed no interaction effect according to different traditional cardiovascular risk factors (<u>Table S1</u>).

The ORs for new-onset PAD for the various combinations of Hcy and traditional risk factors are displayed in Table 3 and Figure 1. Assessment of the effect of high Hcy in combination with smoking revealed a joint effect. Compared with non-smoking participants with normal Hcy concentration, the OR for incident PAD increased by a factor of 2.08 for participants with high Hey and by a factor of 2.10 for smokers. Smokers with a high Hcy concentration had an OR for incident PAD of 4.44, corresponding to a multiplicative relationship on the odds scale. Although the association between diabetes and PAD in participants with a Hey concentration <10µmol/L was not significant, there was a significant association of a combination of Hcy concentration >10µmol/L and the presence of diabetes with PAD risk (OR = 3.67, 95% CI: 1.25-10.80, P = 0.018). This effect was stronger than that of Hcy alone in non-diabetic subjects. Also, there were joint effects of high Hcy concentration and aging on the incidence of PAD. Participants with both elevated Hcy levels and older age had the highest risk of new-onset PAD (OR = 4.28, 95% CI: 1.83–10.01, P < 0.001).

Among the four subgroups, hypertensive patients with hyperhomocysteinemia had the highest risk of incident PAD but were not statistically significant compared with non-hypertensives with normal Hcy concentration (OR = 1.99, 95% CI: 0.75-5.27, P = 0.166). However, there

seemed to be a trend towards increased risk across the four subgroups (*P* for trend=0.026). The joint effect of high Hcy concentration and dyslipidemia on new-onset PAD also existed (*P* for trend=0.035). With regard to the joint effect of Hcy and sex, there was also a trend towards increased risk across four different groups (*P* for trend=0.016), females with hyperhomocysteinemia had the highest risk of incident PAD but was not statistically significant compared with males with Hcy<10 μ mol/L (OR = 2.79, 95% CI: 0.35–22.11, *P* = 0.330). However, there seemed no joint effect of high Hcy concentration and BMI.

Discussion

The major findings of the present study were as follows. A plasma Hcy concentration of $\geq 10 \mu mol/L$ was independently associated with a higher incidence of PAD in a community-based cohort in China. Although participants with higher plasma Hcy concentration tended to be older and have more traditional risk factors for CVD, such as hypertension and a smoking habit, than those with lower concentrations, the association remained significant after adjusting for these risk factors. Furthermore, high Hcy concentration and traditional cardiovascular risk factors such as smoking, diabetes and aging had a joint effect on the incidence of PAD. Our results extend previous findings and suggest that hyperhomocysteinemia and some traditional risk factors have joint effects on PAD risk. Thus, the assessment of plasma Hcy concentration alongside

Subgroups		Incidence, n(%)	OR (95% CI) ^a	P for effect	P for trend
Hcy × Hypertension					0.026
Hcy<10µmol/L	Hypertension=no	7 (1.56)	Ref.		
Hcy<10µmol/L	Hypertension=yes	5 (1.46)	0.68 (0.19-2.44)	0.553	
Hcy≥I0µmol/L	Hypertension=no	32 (2.87)	1.52 (0.64–3.57)	0.343	
Hcy≥I0µmol/L	Hypertension=yes	54 (4.45)	1.99 (0.75–5.27)	0.166	
Hcy × Diabetes					0.020
Hcy<10µmol/L	Diabetes=no	5 (0.82)	Ref.		
Hcy<10µmol/L	Diabetes=yes	7 (3.91)	2.96 (0.83-10.55)	0.094	
Hcy≥I0µmol/L	Diabetes=no	56 (3.18)	3.20 (1.23-8.31)	0.017	
Hcy≥I0µmol/L	Diabetes=yes	30 (5.31)	3.67 (1.25–10.80)	0.018	
Hcy × Dyslipidemia					0.035
Hcy<10µmol/L	Dyslipidemia=no	l (0.45)	Ref.		
Hcy<10µmol/L	Dyslipidemia=yes	(1.93)	3.28 (0.41–26.32)	0.263	
Hcy≥I0µmol/L	Dyslipidemia=no	18 (2.78)	5.83 (0.75-45.49)	0.092	
Hcy≥I0µmol/L	Dyslipidemia=yes	68 (4.05)	5.72 (0.76-42.85)	0.090	
Hcy × Smoking					0.003
Hcy<10µmol/L	Smoking=no	(1.49)	Ref.		
Hcy<10µmol/L	Smoking=yes	l (l.85)	2.10 (0.25–17.44)	0.494	
Hcy≥I0µmol/L	Smoking=no	64 (3.58)	2.08 (1.04-4.14)	0.038	
Hcy≥10µmol/L	Smoking=yes	22 (4.10)	4.44 (1.77–11.12)	0.001	
Hcy × BMI					
Hcy<10µmol/L	BMI<24 kg/m ²	I (0.4I)	Ref.		0.077
Hcy<10µmol/L	24≤BMI<28 kg/m ²	6 (1.67)	4.67 (0.55–39.70)	0.158	
Hcy<10µmol/L	BMI≥28 kg/m ²	5 (2.66)	4.56 (0.51-40.52)	0.174	
Hcy≥I0µmol/L	BMI<24 kg/m ²	25 (4.20)	9.33 (1.22–71.16)	0.031	
Hcy≥I0µmol/L	24≤BMI<28 kg/m ²	35 (3.09)	6.45 (0.86-48.54)	0.071	
Hcy≥I0µmol/L	BMI≥28 kg/m ²	26 (4.34)	7.18 (0.94–54.74)	0.057	
Hcy × Age					<0.001
Hcy<10µmol/L	Age<60 years	8 (1.23)	Ref.		
Hcy<10µmol/L	Age≥60 years	4 (2.88)	1.87 (0.52-6.69)	0.334	
Hcy≥I0µmol/L	Age<60 years	37 (2.49)	2.07 (0.93-4.62)	0.076	
Hcy≥10µmol/L	Age≥60 years	49 (5.83)	4.28 (1.83–10.01)	<0.001	
Hcy × Sex					0.016
Hcy<10µmol/L	Male	I (I.20)	Ref.		
Hcy<10µmol/L	Female	(1.55)	1.32 (0.16–11.15)	0.797	
Hcy≥I0µmol/L	Male	32 (2.98)	1.80 (0.23–13.78)	0.572	
Hcy≥10µmol/L	Female	54 (4.31)	2.79 (0.35–22.11)	0.330	

Table 3 Odds Ratio for	or New-Onset	PAD Associated	d with the	e Various	Combinations	of High	Hcy	Concentration	and	Traditional
Cardiovascular Risk Fac	ctors									

Notes: ^aAdjusted for baseline ABI, sex, age, body mass index, estimated glomerular filtration rate subgroup, current smoking and drinking, hypertension, diabetes, dyslipidemia, cardiovascular disease, and the use of antihypertensive, lipid-lowering, or hypoglycemic agents. (When the odds ratio of new-onset PAD associated with the combination of high Hcy concentration and a certain factor is assessed, this factor is not adjusted. For example, dyslipidemia is not adjusted when the odds ratio of new-onset PAD associated with the combination of high Hcy concentration and dyslipidemia is assessed).

Abbreviations: Hcy, homocysteine; OR, odds ratio; CI, confidence interval; PAD, peripheral artery disease; ABI, ankle brachial index; BMI, body mass index.

traditional cardiovascular risk factors may improve PAD risk stratification.

The median plasma Hcy concentration at baseline was similar to that reported from previous large-scale studies conducted in Chinese populations (median 12.5 μ mol/L, IQR 10.5–15.5 μ mol/L),²⁷ but was higher than that reported from the Framingham study²⁸ and those reported in other countries.²⁹ This might be attributable to the high



Figure I Odds ratio for new-onset PAD associated with the combinations of high Hcy concentration and hypertension (**A**), diabetes (**B**), dyslipidemia (**C**), smoking (**D**), BMI (**E**), Age (**F**) and Sex (**G**)^a.

Notes: ^aAdjusted for baseline ABI, sex, age, body mass index, estimated glomerular filtration rate subgroup, current smoking and drinking, hypertension, diabetes, dyslipidemia, cardiovascular disease, and the use of antihypertensive, lipid-lowering, or hypoglycemic agents. (When the odds ratio of new-onset PAD associated with the combination of high Hcy concentration and a certain factor is assessed, this factor is not adjusted. For example, dyslipidemia is not adjusted when the odds ratio of new-onset PAD associated with the combination of high Hcy concentration and dyslipidemia is assessed).

Abbreviations: Hcy, homocysteine; OR, odds ratio; PAD, peripheral artery disease; ABI, ankle brachial index; BMI, body mass index.

frequency of the 677TT genotype of the methylene tetrahydrofolate reductase (*MTHFR*) gene in the Chinese population^{30–32} and the different diets and nutritional status in China.³³ Therefore, it is extremely important to focus on the role of Hcy in the Chinese population. Previous studies have suggested that individuals with an Hcy $\geq 10\mu$ mol/L are at a higher risk of cardio- and cerebrovascular events, both in China³⁴ and the U.S.³⁵ Therefore, and in accordance with previous reports,²⁴ we selected 10µmol/L as the cut-off value when assessing the impact of Hcy concentration in the present study.

In 1995, a meta-analysis of three case–control studies conducted by Boushey et al showed that Hcy was a risk factor for PAD (OR=6.8, 95% CI 2.9–15.8).¹⁷ Then, in 2009, another meta-analysis of 14 case–control studies demonstrated that Hcy concentration in patients with PAD was 4.31µmol/L higher than that of controls without PAD.¹⁸ A recent case–control study conducted in a Chinese population also indicated that Hcy was associated with PAD, independent of classical vascular risk factors.²⁰ However, prospective studies have yielded conflicting results. Ridker et al performed a nested case–control study cohort, but no association was found between Hcy and PAD.¹² In 2006, Aboyans et al investigated the

relationship between Hcy concentration and PAD progression, but found no association.¹³ Finally, Monica et al conducted a nested case-control study that showed that Hey concentration was positively associated with the risk of PAD in men, but not in women.¹⁵ In the present study, we followed a cohort of individuals who did not have PAD at baseline, and have shown that high plasma Hcy concentration is an independent risk factor for new-onset PAD. The main reason why our results are inconsistent with previous cohort studies may be the population differences. In the majority of Western populations, folic acid fortification was implemented and their Hcy levels were lower than that of the Chinese population.²⁸ The relative lower Hey level may lead to a negative result in their studies.¹² Laboratory research has shown that Hcy is involved in various pathological processes, such as endothelial dysfunction, oxidative stress, and vascular remodeling,³³ and these mechanisms may explain how Hcy increases the risk of PAD. Since the deleterious effect of Hcy on cardio- and cerebrovascular diseases is now well established, our findings have further elucidated its effect on PAD risk.

Current smoking was found to be associated with an increased risk of new-onset PAD, which was consistent with previous studies.³ However, the definition of current smoking was based on self-reported smoking habits,

measurement of plasma cotinine levels might be needed in the future studies to reflect the smoking status more precisely. Perhaps surprisingly, the effect of diabetes on PAD risk was not significant in the present study. As the outcome of our study was new-onset PAD, subjects with PAD at baseline were excluded. The mechanism of diabetes causing PAD lies in endothelial dysfunction with denervation of the vascular smooth muscle cells, which is a relatively long process. Moreover, the duration of diabetes was found to be associated with the risk of developing PAD, so it might be difficult to show a significant effect of diabetes within 2 to 3 years.³⁶ Hypertension and dyslipidemia on PAD risk were also not significant. Although PAD is associated with similar risk factors to other atherosclerotic diseases, the effects of hypertension and dyslipidemia on PAD are modest.³⁷ A study of 7058 people of \geq 40 years of age from NHANES showed that the OR for PAD associated with hypertension was 1.5 (95% CI: 0.9–2.2, P = 0.05).³⁸ Another study conducted in a multi-ethnic Asian population comprising 4132 participants demonstrated that those with PAD had higher SBP, but the association was lost in multivariate analysis. Furthermore, there was no association between dyslipidemia and PAD.³⁹ It was established that the incidence of PAD increased with aging,³⁷ which was consistent with our results. The effect of sex on PAD has been controversial, as previous evidence showed an increased prevalence of PAD in male, but some recent studies found PAD was more common in female.40,41 Recent global estimates found prevalence of PAD to be the same among gender in high-income countries, but more prevalent in women in low- or middle-income countries like China.³ In our study, no significant association was observed between sex and incident PAD but the trend towards the same direction.

Previous studies have shown that traditional risk factors have joint effects on the risk of PAD when present in combination. Joosten et al performed a prospective study of 44,985 men, with a median follow-up period of 24.2 years, and identified joint effects of combinations of a smoking habit, hypertension, hypercholesterolemia, and type 2 diabetes.⁴² Furthermore, Eraso et al used data from the National Health and Nutrition Examination Survey (NHANES) and included hypertension, diabetes, chronic kidney disease, and smoking as PAD risk factors. The risk of PAD increased with each additional risk factor present, from a non-significant 1.5-fold increase (OR=1.5, 95% CI: 0.9-2.6) in the presence of one risk factor, to a > 10-fold increase (OR=10.2, 95% CI: 6.4-16.3) in the presence of

three or more risk factors.³⁸ However, until now, the effects of combinations of hyperhomocysteinemia and traditional risk factors on PAD risk were largely unknown, our results filled this gap. Our study suggests that special attention is needed in hyperhomocysteinemia in subjects with traditional risk factors such as smokers, elder and patients with diabetes, since they are more likely to suffer from PAD in the future. Early screening and appropriate treatments are essential.

The present study had several limitations. First, study subjects were recruited from a single community in Beijing; therefore, the data might not be representative of populations in other locations in China. The population differences in plasma Hcy concentration might make it necessary to be cautious when generalizing our findings to other populations. Second, the time scale for the development of PAD should be determined more accurately, because we only followed up the participants at two time points. Third, because of the relatively short follow-up period, the incidence of new-onset PAD in the sample was relatively low, which may have affected the results of the subgroup analyses. Therefore, larger studies with longer follow-up times should be conducted to verify the effect of plasma Hcy concentration on PAD and its relationships with other traditional cardiovascular risk factors, which may improve risk prediction and assist with the identification of groups at high risk of PAD in the future.

To our knowledge, this is the largest longitudinal study to date to demonstrate that hyperhomocysteinemia is a significant predictor of PAD in a Chinese population. Because the distribution of genetic polymorphisms of Hcy metabolic enzymes and dietary pattern differs geographically, it is important to provide evidence for distinct populations, including the Chinese population. The present cohort study is also the first to investigate the joint effects of hyperhomocysteinemia and conventional cardiovascular risk factors on PAD risk in China. Although the relatively short follow-up period was a limitation, the present study provides new insight into integrating the effect of hyperhomocysteinemia into certain traditional risk factors on PAD risk.

Conclusions

In conclusion, high plasma Hcy independently predicted new-onset PAD in a Chinese community-based population. There is a joint effect of high Hcy concentration and traditional cardiovascular risk factors such as smoking, diabetes and aging on the incidence of PAD. Therefore, early screening and appropriate treatments are crucial among subjects with traditional risk factors when accompanying with hyperhomocysteinemia.

Abbreviations

Hcy, homocysteine; PAD, peripheral arterial disease; ABI, ankle brachial index; BMI, body mass index; BP, blood pressure; SBP, systolic blood pressure; DBP, diastolic blood pressure; FBG, fasting blood glucose; OGTT, oral glucose tolerance testing; TG, triglyceride; TC, total cholesterol; LDL-C, low-density lipoprotein-cholesterol; HDL-C, highdensity lipoprotein-cholesterol; CVD, cardiovascular disease; TIA, transient ischemic attack; SD, standard deviation; eGFR, estimated glomerular filtration rate; IQR, interquartile range; OR, odds ratio; CI, confidence interval; MTHFR, methylene tetrahydrofolate reductase; NHANES, National Health and Nutrition Examination Survey.

Data Sharing Statement

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request. The datasets intended for sharing have been deidentified.

Ethics Approval and Informed Consent

This study was approved by the ethics committee of Peking University (approval numbers: IRB00001052-11086), and written informed consent was obtained from each participant. The study was conducted in accordance with the ethical guidelines of the Declaration of Helsinki.

Acknowledgments

We are very grateful to the staff of the Gucheng and Pingguoyuan Community Health Centers and the research coordinators who participated in this cohort study.

Author Contributions

All authors reviewed and approved the manuscript, and they agree to be accountable for all aspects of the work. All authors made substantial contributions to conception and design, acquisition of data, or analysis and interpretation of data; took part in drafting the article or revising it critically for important intellectual content; agreed on the journal to which the article will be submitted; gave final approval of the version to be published; and agree to be accountable for all aspects of the work.

Funding

This study was supported by grants from Projects of National Natural Science Foundation of China (grant 81703288), Peking University Medicine Seed Fund for Interdisciplinary Research and The Fundamental Research Funds for the Central Universities (BMU2018MX002), Scientific Research Seed Fund of Peking University First Hospital (2018SF003 and 2018SF071), Key Laboratory of Molecular Cardiovascular Sciences (Peking University), Ministry of Education and NHC Key Laboratory of Cardiovascular Molecular Biology and Regulatory Peptides.

Disclosure

The authors declare no conflicts of interest for this work.

References

- Kullo IJ, Solomon CG, Rooke TW. Peripheral artery disease. N Engl J Med. 2016;374(9):861–871. doi:10.1056/NEJMcp1507631
- Gerhard-Herman MD, Gornik HL, Barrett C, et al. 2016 AHA/ACC guideline on the management of patients with lower extremity peripheral artery disease: a report of the American College of Cardiology/American Heart Association task force on clinical practice guidelines. *Circulation*. 2017;135(12):e726–e779. doi:10.1161/ CIR.000000000000471
- Fowkes FGR, Rudan D, Rudan I, et al. Comparison of global estimates of prevalence and risk factors for peripheral artery disease in 2000 and 2010: a systematic review and analysis. *Lancet.* 2013;382 (9901):1329–1340. doi:10.1016/S0140-6736(13)61249-0
- Hamburg NM, Creager MA. Pathophysiology of intermittent claudication in peripheral artery disease. *Circ J.* 2017;81(3):281–289. doi:10.1253/circj.CJ-16-1286
- Song P, Rudan D, Zhu Y, et al. Global, regional, and national prevalence and risk factors for peripheral artery disease in 2015: an updated systematic review and analysis. *Lancet Glob Health*. 2019;7 (8):e1020–e1030. doi:10.1016/S2214-109X(19)30255-4
- 6. Vos T, Allen C, Arora M, et al. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990–2015: a systematic analysis for the global burden of disease study 2015. *Lancet*. 2016;388(10053):1545–1602.
- Wang H, Naghavi M, Allen C, et al. Global, regional, and national life expectancy, all-cause mortality, and cause-specific mortality for 249 causes of death, 1980–2015: a systematic analysis for the global burden of disease study 2015. *Lancet*. 2016;388(10053):1459–1544.
- Brostow DP, Hirsch AT, Collins TC, Kurzer MS. The role of nutrition and body composition in peripheral arterial disease. *Nat Rev Cardiol.* 2012;9(11):634–643. doi:10.1038/nrcardio.2012.117
- McCully KS. Homocysteine and the pathogenesis of atherosclerosis. *Expert Rev Clin Pharmacol.* 2015;8(2):211–219. doi:10.1586/ 17512433.2015.1010516
- Malinow MR. Plasma homocyst(e)ine and arterial occlusive diseases: a mini-review. *Clin Chem.* 1995;41(1):173–176. doi:10.1093/clinchem/41.1.173
- McCully KS. Homocysteine metabolism, atherosclerosis, and diseases of aging. *Compr Physiol*. 2015;6(1):471–505.
- Ridker PM, Stampfer MJ, Rifai N. Novel risk factors for systemic atherosclerosis: a comparison of C-reactive protein, fibrinogen, homocysteine, lipoprotein(a), and standard cholesterol screening as predictors of peripheral arterial disease. *JAMA*. 2001;285 (19):2481–2485. doi:10.1001/jama.285.19.2481

- Aboyans V, Criqui MH, Denenberg JO, Knoke JD, Ridker PM, Fronek A. Risk factors for progression of peripheral arterial disease in large and small vessels. *Circulation*. 2006;113(22):2623–2629. doi:10.1161/CIRCULATIONAHA.105.608679
- 14. Taylor LM, DeFrang RD, Harris EJ, Porter JM. The association of elevated plasma homocyst(e)ine with progression of symptomatic peripheral arterial disease. J Vasc Surg. 1991;13(1):128–136. doi:10.1016/0741-5214(91)90020-U
- Bertoia ML, Pai JK, Cooke JP, et al. Plasma homocysteine, dietary B vitamins, betaine, and choline and risk of peripheral artery disease. *Atherosclerosis*. 2014;235(1):94–101. doi:10.1016/j.atherosclerosis.20 14.04.010
- Pradhan AD, Shrivastava S, Cook NR, Rifai N, Creager MA, Ridker PM. Symptomatic peripheral arterial disease in women. *Circulation*. 2008;117 (6):823–831. doi:10.1161/CIRCULATIONAHA.107.719369
- Boushey CJ, Beresford SA, Omenn GS, Motulsky AG. A quantitative assessment of plasma homocysteine as a risk factor for vascular disease. Probable benefits of increasing folic acid intakes. *JAMA*. 1995;274(13):1049–1057. doi:10.1001/jama.1995.03530130055028
- Khandanpour N, Loke YK, Meyer FJ, Jennings B, Armon MP. Homocysteine and peripheral arterial disease: systematic review and meta-analysis. *Eur J Vasc Endovasc Surg.* 2009;38(3):316–322. doi:10.1016/j.ejvs.2009.05.007
- Cheng SW, Ting AC, Wong J. Fasting total plasma homocysteine and atherosclerotic peripheral vascular disease. *Ann Vasc Surg.* 1997;11 (3):217–223. doi:10.1007/s100169900037
- Rong D, Liu J, Jia X, et al. Hyperhomocysteinaemia is an independent risk factor for peripheral arterial disease in a Chinese Han population. *Atherosclerosis*. 2017;263:205–210. doi:10.1016/j.atherosclerosis.20 17.05.006
- Fan F, Qi L, Jia J, et al. Noninvasive central systolic blood pressure is more strongly related to kidney function decline than peripheral systolic blood pressure in a Chinese community-based population. *Hypertension*. 2016;67(6):1166–1172. doi:10.1161/HYPERTENSIONAHA.115.07019
- 22. Wang Y, Li X, Qin X, et al. Prevalence of hyperhomocysteinaemia and its major determinants in rural Chinese hypertensive patients aged 45–75 years. Br J Nutr. 2012;109(7):1284–1293. doi:10.1017/ S0007114512003157
- Qin X, Huo Y. H-Type hypertension, stroke and diabetes in China: opportunities for primary prevention. *J Diabetes*. 2016;8(1):38–40. doi:10.1111/1753-0407.12333
- 24. Stanger O, Herrmann W, Pietrzik K, et al. DACH-LIGA homocystein (German, Austrian and Swiss Homocysteine Society): consensus paper on the rational clinical use of homocysteine, folic acid and B-vitamins in cardiovascular and thrombotic diseases: guidelines and recommendations. *Clin Chem Lab Med.* 2003;41(11):1392–1403. doi:10.1515/CCLM.2003.214
- McCully KS. Homocysteine, vitamins, and vascular disease prevention. *Am J Clin Nutr.* 2007;86(5):1563S–1568S. doi:10.1093/ajcn/86.5.1563S
- 26. Momin M, Fan F, Li J, et al. Associations of plasma homocysteine levels with peripheral systolic blood pressure and noninvasive central systolic blood pressure in a community-based Chinese population. *Sci Rep.* 2017;7(1):6316. doi:10.1038/s41598-017-06611-3
- Huo Y, Li J, Qin X, et al. Efficacy of folic acid therapy in primary prevention of stroke among adults with hypertension in China: the CSPPT randomized clinical trial. *JAMA*. 2015;313(13):1325–1335. doi:10.1001/jama.2015.2274
- 28. Jacques PF, Bostom AG, Wilson PW, Rich S, Rosenberg IH, Selhub J. Determinants of plasma total homocysteine concentration in the Framingham offspring cohort. *Am J Clin Nutr.* 2001;73 (3):613–621. doi:10.1093/ajcn/73.3.613

- Akanji AO, Thalib L, Al-Isa AN. Folate, vitamin B₁₂ and total homocysteine levels in Arab adolescent subjects: reference ranges and potential determinants. *Nutr Metab Cardiovasc Dis.* 2012;22 (10):900–906. doi:10.1016/j.numecd.2010.10.020
- Xu X, Li J, Sheng W, Liu L. Meta-analysis of genetic studies from journals published in China of ischemic stroke in the Han Chinese population. *Cerebrovasc Dis.* 2008;26(1):48–62. doi:10.1159/000135653
- Kang SS, Wong PW, Bock HG, Horwitz A, Grix A. Intermediate hyperhomocysteinemia resulting from compound heterozygosity of methylenetetrahydrofolate reductase mutations. *Am J Hum Genet*. 1991;48(3):546–551.
- Frosst P, Blom HJ, Milos R, et al. A candidate genetic risk factor for vascular disease: a common mutation in methylenetetrahydrofolate reductase. *Nat Genet*. 1995;10(1):111–113. doi:10.1038/ng0595-111
- Zaric BL, Obradovic M, Bajic V, Haidara MA, Jovanovic M, Isenovic ER. Homocysteine and hyperhomocysteinaemia. *Curr Med Chem*. 2019;26 (16):2948–2961. doi:10.2174/0929867325666180313105949
- 34. Sun Y, Chien K-L, Hsu H-C, Su T-C, Chen M-F, Lee Y-T. Use of serum homocysteine to predict stroke, coronary heart disease and death in ethnic Chinese. 12-year prospective cohort study. *Circ J*. 2009;73(8):1423–1430. doi:10.1253/circj.CJ-08-1077
- 35. Towfighi A, Markovic D, Ovbiagele B. Pronounced association of elevated serum homocysteine with stroke in subgroups of individuals: a nationwide study. *J Neurol Sci.* 2010;298(1–2):153–157. doi:10.10 16/j.jns.2010.07.013
- 36. Al-Delaimy WK, Merchant AT, Rimm EB, Willett WC, Stampfer MJ, Hu FB. Effect of type 2 diabetes and its duration on the risk of peripheral arterial disease among men. *Am J Med.* 2004;116 (4):236–240. doi:10.1016/j.amjmed.2003.09.038
- 37. Aboyans V, Ricco JB, Bartelink MEL, et al. 2017 ESC guidelines on the diagnosis and treatment of peripheral arterial diseases, in collaboration with the European Society for Vascular Surgery (ESVS): document covering atherosclerotic disease of extracranial carotid and vertebral, mesenteric, renal, upper and lower extremity arteriesEndorsed by: the European Stroke Organization (ESO)the task force for the diagnosis and treatment of peripheral arterial diseases of the European Society of Cardiology (ESC) and of the European Society for Vascular Surgery (ESVS). *Eur Heart J.* 2018;39(9):763–816. doi:10.1093/eurheartj/ehx095
- 38. Eraso LH, Fukaya E, Mohler ER 3rd, Xie D, Sha D, Berger JS. Peripheral arterial disease, prevalence and cumulative risk factor profile analysis. *Eur J Prev Cardiol.* 2014;21(6):704–711. doi:10.11 77/2047487312452968
- 39. Subramaniam T, Nang EEK, Su Chi L, et al. Distribution of ankle brachial index and the risk factors of peripheral artery disease in a multi-ethnic Asian population. *Vasc Med.* 2011;16(2):87–95. doi:10.1177/1358863X11400781
- Norgren L, Hiatt WR, Dormandy JA, et al. Inter-society consensus for the management of peripheral arterial disease (TASC II). J Vasc Surg. 2007;45(Suppl S):S5–S67. doi:10.1016/j.jvs.2006.12.037
- Patel T, Baydoun H, Patel NK, et al. Peripheral arterial disease in women: the gender effect. *Cardiovasc Revasc Med.* 2020;21 (3):404–408. doi:10.1016/j.carrev.2019.05.026
- Joosten MM, Pai JK, Bertoia ML, et al. Associations between conventional cardiovascular risk factors and risk of peripheral artery disease in men. JAMA. 2012;308(16):1660-1667. doi:10.1001/jama.2012.13415

Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy

Dovepress

Publish your work in this journal

Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy is an international, peer-reviewed open-access journal committed to the rapid publication of the latest laboratory and clinical findings in the fields of diabetes, metabolic syndrome and obesity research. Original research, review, case reports, hypothesis formation, expert opinion and commentaries are all considered for publication. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/diabetes-metabolic-syndrome-and-obesity-targets-and-therapy-journal